Author's response to reviews

Title: Maternal Associated Factors of Low Birth weight: A Hospital Based Cross-sectional mixed Study in Tigray, Northern Ethiopia

Authors:

Meresa Weldu Gebremedhin (mere2010ec@yahoo.com)
Fentie Getahun Ambaw (fentambaw@yahoo.com)
Eleni Mersha Admassu (eleniam16@gmail.com)
Haileselasie Alema Berhane (haileselasieb@yahoo.com)

Version: 2 Date: 25 September 2014

Author's response to reviews:

Dear all

BioMed Central Public Health journal (BMC) is our choice of publication because of its reputability and articles published in the journal are available to a world wide range. The promotion of authors from developing countries is other reason that made this journal favorable for publication. Furthermore the copyright of articles are well secured legally and the progress of publication can be followed via email updates followed by fast publication after acceptance.

With regards

Meresa, Corresponding author